

Review



A Comprehensive Review of Small-Signal Stability and Power Oscillation Damping through Photovoltaic Inverters

Nikolay Nikolaev 🔍, Kiril Dimitrov and Yulian Rangelov *

Electric Power Engineering Department, Technical University of Varna, Studentska str. 1, 9010 Varna, Bulgaria; n.nikolaev@tu-varna.bg (N.N.); k.dimitrov55@tu-varna.bg (K.D.)

* Correspondence: y.rangelov@tu-varna.bg; Tel.: +359-899328086

Abstract: This paper focuses on the methods that ensure the rotor angle stability of electric power systems, which is most frequently analyzed with small-signal models. Over the past several decades, power system stabilizers (PSSs) for conventional excitation systems were the main tools for improving the small-signal stability of electromechanical oscillatory modes. In the last decade, power oscillation damping (POD) control implemented in photovoltaic (PV) inverters has been considered an alternative to PSSs. As PV generation undergoes massive rollout due to policy directions and renewable energy source integration activities, it could potentially be used as a source of damping, which is crucial for sustaining the rotor angle stability of the remaining in-service synchronous generators. Several studies have already been dedicated to the development of different damping strategies. This paper contributes to the existing research in power system stability by providing a comprehensive review of the effects of PV generation on small-signal stability, as well as the recent evolution of POD control through PV inverters. The features and impacts of the various ways to realize POD controllers are assessed and summarized in this paper. Currently, detailed information and discussions on the practical application of PV inverter PODs are not available. This paper is, thus, intended to initiate a relevant discussion and propose possible implementation approaches concerning the topic under study.

Keywords: power oscillation damping; small-signal stability; photovoltaic inverter; electromechanical oscillation; virtual synchronous generator; microgrid

1. Introduction

Large interconnected power systems offer several advantages such as liberalized cross-border markets, shared power reserves, and reliable power supply. One of the major challenges encountered by such big power systems relates to the maintenance of the stability and reliability of their operation. Definitions and classifications of the phenomena of power system stability are available in [1,2]. These definitions suggest that, upon a disturbance, to ensure the stability of electric power systems, a transition to another steady state is required while maintaining a non-interruptible power delivery. The stability should be preserved in any possible operating conditions of the power system, including the constantly changing generation, demand, and weather conditions. According to [1], there are three main groups of power system stability" rotor angle stability, frequency stability, and *voltage stability*. This paper focuses primarily on the methods that ensure rotor angle stability, which is most frequently analyzed using small-signal models. For the past several decades, the power system stabilizers (PSSs) applied in the excitation systems of conventional generating units were the main tools used for influencing the electromechanical oscillatory modes of the power system, which also improved its small-signal stability. It is a well-known fact that these oscillations have low frequencies and can be classified into two main categories: local (typically 0.7–2 Hz) and interarea (0.1–0.7 Hz). The first recorded event of low-frequency oscillations happened in the 1960s [3]. Electromechanical



Citation: Nikolaev, N.; Dimitrov, K.; Rangelov, Y. A Comprehensive Review of Small-Signal Stability and Power Oscillation Damping through Photovoltaic Inverters. *Energies* 2021, 14, 7372. https://doi.org/10.3390/ en14217372

Academic Editors: Emilio Lorenzani and Abu-Siada Ahmed

Received: 6 October 2021 Accepted: 1 November 2021 Published: 5 November 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). oscillations are inherent to the synchronous generators and their rotors can swing either with or against the rotors of the other machines in the system. The oscillations can also be classified as *single-mode power oscillations* when there is only one resonant frequency or as *multi-mode power oscillations* when many dominant frequencies are superimposed (for instance, both local and interarea).

Several cases have been documented in which interarea oscillations have led to blackouts of power systems due to degraded small-signal stability [4–9]. Several existing publications analyzed the nature of interarea oscillations and the possible measures to remedy their impact [9–15]. As an alternative to PSSs, some studies outlined the possibility of using the converters of FACTS and HVDC to mitigate stability issues [15–17]. Another large body of research studies, especially in the past 10 years, has focused on the application of power oscillation damping (POD) control implemented in photovoltaic (PV) inverters. This paper aims to provide a comprehensive review of the utilization of PV inverters for influencing the small-signal stability of power systems.

The nature of the electromechanical oscillations is strongly related to the rotating masses of the synchronous generators and the kinetic energy they store, i.e., their inertia. Recent changes observed in the generation portfolio of many power systems involve the gradual substitution of synchronous machines with the inertia-less inverter-based generation, which has a significant impact on the power system dynamics. Multiple instances have been recorded where, for short periods, the entire load of a country is covered by renewable generation [18]. Some transmission system operators (TSOs) deal with such situations by limiting the output of PV inverters [19]. Some suggestions posit [20] that such limitations will no longer be needed if better control solutions are designed for PV inverters. The issues concerning stability that arise from the increasing share of renewable generation and the corresponding uncertainties continue to be an important topic in the scientific and professional communities [21–23]. The authors of [22] stressed the need for additional research dedicated to the effects and proportions of grid-following and gridforming inverters. Mathematical models for low-inertia power systems are being developed that afford special attention to the generation control [23]. The issue of decreasing inertia can be partly resolved by applying coordinated control of inverters [21] or by introducing synthetic inertial response [24]. The authors of [25] proposed that the global inertial concept be replaced by a local inertia concept, depending on the ratio between conventional and inverter-based generation at a given time instance. The inertia constants, on the other hand, can be considered as time functions. A comprehensive study of the distribution of inertia in continental Europe, the impact of neighboring areas, and the economic aspects related to the changing system dynamics was presented in [26]. The possible benefits of implementing an inertia monitoring system were also assessed.

Some scientists expressed with certainty that, with suitable control, inertia-less power systems can sustain a high degree of stability under various conditions and radically improve the response of the system to disturbances [27]. An underestimated source of inertia was considered in [28]: the electric motor as a part of complex loads. Its relevance, however, depends on the motor share of the total load.

The other types of stability phenomena are also affected by high penetration levels of PV generation. The voltage stability of a part of the Texas grid was studied in [29]. The analysis was carried out with up to 65% PV penetration. The study revealed that the voltage stability is threatened due to the insufficient volt/var control of PV inverters. An improved volt/var control was developed in [30]. In [31], it was demonstrated that the utilization of inverters' reactive power capability is important for maintaining the voltage stability of power systems.

The frequency stability is also expected to degrade due to the intermittent nature of solar power [32–34]. A good candidate solution for maintaining the power system frequency is the utilization of battery energy storage [35,36]. Operating the inverter as a virtual inertia device is also a viable solution, and a coordinated approach was developed

in [37]. It is later discussed in Section 4.3 of this review that the virtual inertia control of PV generation can also be effectively applied to POD.

The modeling of conventional synchronous generators is crucial for stability analysis. Relevant studies on the topic can be found in the literature [2,38,39].

The literature survey reveals that the transition of power systems to a form of generation that is dominantly inverter-based will likely not be flawless. Taking the conventional synchronous machines out of service compromises the inertial response and excludes the PSS as a main tool for damping electromechanical oscillations. Hence, there is an immediate need for an alternative source of damping control. In the past 15 years, a large number of PV systems were integrated into the power systems, and they could potentially be used as a source of damping, which is crucial to sustaining the rotor angle stability of the remaining in-service synchronous generators. Intensive research efforts have already been aimed at developing various damping strategies, which signals a need for reviewing the different alternatives for POD control from PV generation.

This paper contributes to the study of sustainable integration of renewables by providing a comprehensive review of the effects of PV generation on the small-signal stability, as well as the recent advances in POD control through PV inverters. POD controllers are very diverse. Their pros and cons, features, and properties are summarized and critically reviewed. The practical application of PV inverter PODs is not well established in the extant literature. Here, we initiate a discussion and propose possible approaches for its implementation.

Two very recent publications were brought to the authors' attention after the submission of this review [40,41]. These references are excellent surveys on the application of fractional order control for many different applications, including POD. They demonstrate and conclude that the fractional-order damping controller is more robust than its conventional counterpart, but the tuning methodologies are not yet well established. Hence, their practical application needs further development. While the two surveys focus solely on fractional-order controllers, our review attempts to present a wide range of POD topologies.

The paper is organized as follows: Section 2 briefly describes the basics of PV inverter control. Section 3 presents a review of the impact of PV inverters on the small-signal stability of electric power systems and outlines the main findings. Section 4 delineates the classification of the different POD strategies for PV generation into three groups. Section 5 reviews the application of POD in microgrids. Section 6 summarizes the features of the POD controllers. Section 7 discusses the practical application of damping control, and Section 8 presents the conclusions of the present comprehensive review.

2. Overview of PV Inverter System Control

Inverters are generally classified into single-stage and two-stage inverters. Single-stage inverters, where the PV array is directly fed to the DC/AC inverter, are more efficient than two-stage inverters, which have an additional DC/DC converter [42]. This was confirmed in [43], where a review of 45 different inverter topologies was presented. Some single-stage topologies have a lower number of switches operating with pulse width modulation (PWM). The PV plants in most power system applications are single-stage; hence, they are considered in this section.

The electronic power converters can be classified in terms of the main electrical variables. Two topologies are used in power conversion applications: voltage-source inverter (VSI) and current-source inverter (CSI). The CSI uses thyristors and a large series inductor connected at the DC side. The VSI uses IGBT or MOSFET transistors, and the DC voltage is supported by a parallel capacitor. Both these types have their respective advantages and disadvantages, which were summarized in [44,45]. The main advantage of the VSI is that its power control is very flexible, which provides good steady-state performance and excellent response during transient phenomena.

The typical control of a grid-connected VSI is shown in Figure 1 [46]. The inverter is controlled in a dq reference frame aligned to the voltage of the invertor's output. The dq

transformation reduces the three-phase alternating quantities into two static DC quantities. The DC quantities provide easier control and filtering and allow for decoupled active and reactive power control [46–48]. The inverters are connected to the grid via various topologies of passive filters such as *L*, *LC*, or *LCL* [49–51]. In Figure 1, an *L* filter is used to eliminate the higher-frequency current components from the inverter output. L_f denotes the output filter inductance and V^{θ} represents the voltage at the point of common coupling (PCC).



Figure 1. Control of grid-connected inverter.

This paper demonstrates that inverter control has a significant influence on smallsignal stability analysis in power systems with prevailing generation from renewable energy sources and their inverter-based generators, especially on their current control and phase-locked loops (PLLs).

Key guidelines for modeling PV power plants for power system stability studies are available in [52,53]. The different modules that are part of the control of the grid-connected inverter are described below.

2.1. Current Control

Figure 2 illustrates the current control loop. It is the fast inner control loop of the inverter. It presents a dq cross-decoupling current control scheme, where both i_d and i_q are controlled using independent PI controllers [47,54–58]. The resulting output is the inverter's output voltage reference in the dq coordinate system.

2.2. Active and Reactive Power Control

To control the grid-injecting mode, additional blocks (power calculation block and power control block) are needed (see Figure 1). With two input signals, i_{dq} and u_{dq} , the power calculation block computes the active and reactive power (*P* and *Q*). These are compared to the references P_{ref} and Q_{ref} to form the control errors of the power control block. The real power value of the reference is supplied by the maximum power point tracking (MPPT). That power control acts as a slower outer loop, where two independent control blocks (for active and reactive power) generate the corresponding signals for the reference of the current controller [46]. The active and reactive power controllers are shown

in Figure 3. It should be noted that active power is controlled by varying the DC link voltage. Hence, the active power and DC voltage control are all the same modules.



Figure 2. Current control block.



Figure 3. Power control block.

2.3. Pulse Width Modulation

The PWM technique is frequently applied in the design of inverters. It is the most common type of modulation for VSC [44]. There are different types of PWM such as single, multiple, sinusoidal, and modified sinusoidal. The *dq* voltage reference is reverse-transformed into the *ABC* reference frame. The obtained three-phase instantaneous reference voltages are used to produce the PWM signals using triangular waves. The reference transformation utilizes the phase angle provided by the PLL to ensure synchronization to the grid.

2.4. Phase-Locked Loop

The dq frame is aligned to the PCC voltage phasor so that the inverter can synchronize with the grid. This task is performed by a PLL [52,59], as depicted in Figure 4. The phase angle θ_{ref} is detected by synchronizing the reference frame with the vector of the three-phase voltage. The dq frame can be considered synchronized when u_q is equal to the desired value $u_{q,des}$ (zero) [46]. It is to be noted that, if $u_{q,des}$ is nonzero, the dq frame will be misaligned.



Figure 4. Phase-locked loop.

3. Small-Signal Stability of Power Systems with PV Generation

The effects of PV generation on the small-signal stability of power systems have been studied for more than a decade now. Initially, the research was conducted on simplified test models with up to a few machines. Mostly, either the single-machine infinite bus (SMIB) system whose transmission line is bisected to couple the PV inverter or the well-known Kundur's four-machine test model was used for the research. Then, large-scale studies were performed on models that represented real power systems. As detailed later in this paper, different methods were employed to analyze the oscillation modes and damping of the power systems under study. The results appear to be controversial and are not conclusive.

The stability of a SMIB model with PV generation placed in the middle of the transmission tie was studied in [60]. Using damping torque computations and dynamic simulation techniques, the authors demonstrated that the PV generation may have either a positive or a detrimental impact on the small-signal and dynamic stability of the system, depending on its operating conditions. The damping torque produced by the inverters changes its sign with the power system's varying operating points. The study presented in [61] arrived at the same conclusions. This shows that the closer the PV generation is to the synchronous machine, the greater is the damping it induces.

The authors of [62] presented small-signal stability results from Kundur's fourgenerator test system, where one of the generators was gradually replaced by PV generation. Contrary to the findings of the previous research, this study indicates an improvement of the damping when the PV share is increased and a slight degradation of the damping when the synchronous machine is fully disconnected. The authors did not comment upon this deviation or presented an analysis to explain why their results were different from those of preceding studies. Similar conclusions were drawn in [63] on the basis of both dynamic simulations and modal analysis.

The topology of the inverter also affects the damping ability provided by PV generation [64]. In most cases, voltage-source converters perform better than current-source converters in terms of small-signal stability. It was also found that the location and size of the PV injection influence the damping of the critical modes. The same authors demonstrated how PV installation equipped with either an ultra-capacitor or battery storage is capable of improving the oscillation damping in a multimachine infinite bus setup [65].

A bigger (14-bus) test system was studied in [66]. The setup comprised three synchronous machines and a PV generator. The impact of factors such as solar irradiation, cell temperature, and load on the electromechanical oscillation damping were assessed. The authors concluded that the damping is not sensitive to cell temperature, while a positive correlation was identified between the damping and the irradiance.

Shah et al. studied how the degree of PV generation dispersity affects the small-signal stability of bulk power systems [67]. PV penetration levels of up to 15% were analyzed in this study. The oscillation frequency was observed to increase slightly, while the damping reduced. This conclusion was based solely on analyzing one electromechanical mode, which may not be conclusive regarding the overall damping. It was demonstrated that the

trend of stability degradation persists, irrespective of the load model: constant Z, I, or P. They concluded that a more dispersed generation resulted in better damping. Furthermore, the power factor control performs better than the voltage control in terms of power system stability. Another review of wind power generation found that, while power factor control enhances the damping in some cases, it degrades the damping in other instances [68]. We did not find other papers dedicated to the impact of PV generation control mode on electromechanical damping. Therefore, the conclusion that the power factor control is beneficial shall be revised as new research becomes available.

In 2014, a comprehensive study of the small-signal stability of western North America's power system was published [69]. An increase of up to 15% of the share of renewables (both PV and wind) was studied. The publication concluded that, while the modal damping remained unchanged, the oscillation frequency increased slightly. The authors' explanation for the latter is that the overall moment of inertia in the system decreased with the increase in renewable generation. In [70], the stability of the Eastern Interconnected US grid was studied for PV penetration up to 65%. The analysis was based on time-domain simulations and the Matrix Pencil method. This study confirmed that the oscillation frequency increased; however, unlike [69], the results of [70] indicated that the damping degraded slightly as the PV share increased. It is likely that [69] did not detect changes in the damping since the considered inverter generation was just 15%.

Another large-scale study was presented in [71]. The Texas 2000-bus system was tested for PV penetration levels up to 50%. The small-signal analysis indicated that both the electromechanical oscillations frequency and the damping decreased slightly with higher PV injection. The damping trend corresponds with previous research. However, the decreasing oscillation frequency contradicts the evidence of previous studies and is counterintuitive to the reduced inertia in the system. The authors did not explain this observation, and one must consider this result with caution.

An interesting approach for analyzing the power system stability with increasing intermittency from renewables was developed in [72]. The authors proposed the use of a combination of offline- and online trained decision trees to classify and predict the electromechanical modes using the wide-area measurement system (WAMS). The decision tree output can be used to identify proper remedial actions. The offline training requires the simulation of multiple operating scenarios, accounting for the uncertainties of the grid, especially of the renewable generators. The simulations presented in this study suggested a very good identification of the electromechanical modes as compared to the full eigenvalue analysis.

In [73], in addition to the PV generation, the study of stability was extended using a battery energy storage system. It was demonstrated that the battery control can significantly improve the oscillation damping of both the local and the interarea modes. The battery control was equipped with a frequency feedback loop that allows a response to the electromechanical oscillations. The gain of this control induced a great mode shift and could be subject to optimization in terms of maximizing the damping. A slight improvement in damping was also confirmed in [71].

Jia et al. studied the small-signal stability of multiple parallel PV inverters coupled to a weak grid [74]. The authors demonstrated that the gain of the PLL has a significant impact on the system damping. The gain also affects the oscillation frequency. The results were confirmed in [75], where Jia et al. also proposed a virtual inductance control to improve the stability. It was established in [76] that the X/R ratio of the grid impedance significantly affects the stability of the inverter control. The stable operation can be maintained by increasing the LCL filter, controller time delay, and passive damping resistance or by decreasing the proportional controller parameters, i.e., by making it less responsive/sensitive.

Most of the available stability studies were based on deterministic methods. Gurung et al. applied a cumulant method to analyze the small-signal stability by considering the uncertainties of PV generation [77]. This method is based on the combination of sensitivity analysis and Gram–Charlier expansion to produce the probability density function and cumulative density functions of the electromechanical damping. Such data are useful and fundamental for the risk-based reliability analysis of the power system. The study concluded that a higher PV penetration leads to a higher probability of instability. Another probabilistic approach, based on Monte Carlo simulation, was applied in [78] to evaluate the impact of both wind and PV generation. It was concluded that wind generation has a stronger influence on low-frequency oscillations.

The research findings concerning the impact of PV generation on the small-signal stability of power systems can be summarized as follows:

- The frequency of the electromechanical modes generally increases due to the reduced energy stored in the synchronously rotating masses throughout the power system, i.e., due to the reduced inertia. Only one study suggested a decrease in the modal frequency, but it was not supported by sufficient evidence, additional analysis, and explanation.
- A greater share of PV generation leads to a greater impact on the modal damping. At a relatively small percentage of PV generation, the effect on the damping is negligible. With shares above 15%, the damping either decreases or increases more significantly, depending on various factors, as presented in the subsequent bullet points.
- The damping torque induced by the PV inverters changes with the varying of the power system's operating point. This torque could be positive or negative.
- Locating the PV generation near synchronous units has a beneficial impact on the damping.
- Voltage-source converters perform better than current-source converters in terms of small-signal stability.
- The operational parameters of the PV installation have the following impact: The cell temperature does not affect the damping, while the solar irradiance improves it. The latter, however, should be treated with caution as the irradiance strongly correlates to the share of PV generation, and the statement that it improves the damping may not always be true.
- In terms of damping, dispersed PV plants are better than bulk concentrated plants.
- The mathematical model of the load characteristics (constant Z, I, or P) does not have a significant influence on the small-signal stability analysis aimed at studying the PV generation impact.
- The inverter's power factor control performs better than its voltage control in terms of damping. This conclusion is based on a single study, and further research is needed to either confirm or disprove it. It is also possible that a particular type of control cannot be conclusively declared to be superior to another because it may depend on various factors such as grid topology, PV share, power system and operating point.
- The parameters of the inverter's PLL have a strong effect on the damping, especially when coupled to high-impedance grids. This effect can be either positive or negative, depending on the selection of the control parameters.

The inverter-based PV generation changes both the frequency and the damping of the power system electromechanical modes. While the results of research studies regarding the change of modal frequency are more consistent with each other, the studies focused on modal damping indicate that it depends on numerous factors. This makes the analysis of small-signal stability particularly challenging, as factors such as the inverter topology, control scheme, and dispersity must be considered correctly and in a detailed manner to obtain reasonable results. Despite these difficulties, it has become obvious that PV inverters will play a central role in maintaining the power system stability, and dedicated damping control should be implemented.

4. POD in Bulk Power Systems through PV Inverters

4.1. PID, Lead Lag and Other Types of Damping Control

Similar to the studies on small-signal stability, the research approaches that designed and studied POD controllers typically applied the two-area four-generator test system from Kundur [2] and a modified classical SMIB model, where the PV inverter is coupled to the middle of the transmission tie.

The study [79] from 1987 seems to be among the first to identify the need for utilizing the PV inverters for damping power system oscillations. The authors established an inverter control framework that consisted of a power flow model, a swing equation model, and PID controllers. The outputs were the angle and voltage magnitude corrections produced by the PID and D controls. The obtained signals modulated the reference of the inverter's voltage angle and amplitude controllers (Figure 5). The power system frequency and the voltage oscillations were stabilized. The simulations in this study demonstrated the importance of accurate tuning of the POD controller, as it can degrade stability. The authors of [80] adopted a similar approach.



Figure 5. PID control modifying the inverter's voltage magnitude and angle reference.

Francisco de Leon et al. studied the possibility of using unidirectional power control for POD instead of bidirectional control [81]. They successfully demonstrated that this is possible and effective. Furthermore, unidirectional power control is more desirable as it allows the plant to operate at the maximum available power. A rather similar study was also conducted in [82], but the main difference was the use of bidirectional power control achieved by operating the PV system at 50% of the available solar power, thus leaving a range for significant participation in the damping process. The study presented in [83] indicated that slightly better damping is achieved by the bidirectional control as compared to the unidirectional control.

Through eigenvalue sensitivity computations, the authors of [84] proved that the critical mode is much more sensitive to reactive power control from PV than to active power. No previous study demonstrated the effectiveness of reactive overactive power for damping electromechanical oscillations (more recent research studies, however, suggested the contrary [48]). The most suitable stabilizing feedback signals were determined by computing the residues. The voltage magnitude and angle residues were studied. The inverter quadrature current was more sensitive to the voltage magnitude; therefore, it was the authors' choice for stabilizing feedback, although direct current could also be considered an option for other cases. The standard lead-lag structure was employed for the stabilizer. The tuning technique was analytical, but the authors did not elaborate sufficiently on the computation of the required compensation. Lead-lag structure was also adopted in [85–87]. An example of the incorporation of a lead-lag damping control in the inverter's current control loop is depicted in Figure 6, where the POD block consists of a lead-lag compensator and filters. The compensating time constants can be selected analytically, on the basis of the residue method, or using a numerical optimization procedure.



Figure 6. POD modulating the active or reactive current.

A damping controller for PV inverter was designed in [88]. The test system was a simple SMIB with the PV bisecting the transmission line. The feedback signal was the synchronous machine rotor angle deviation. The stabilizer was a simple integrator. Although the authors demonstrated the effectiveness of this approach, it is not very practical as the rotor speed deviation signal is not readily available. In addition, such a signal cannot be easily sent from the conventional power plant to the PV site. The authors extended their work by implementing particle swarm optimization for the POD controller [89].

The authors of [90] proposed the use of a structure-preserving energy function (SPEF) model to derive the POD control structure of PV inverters. Ultimately, the derived controller is a simple filtered derivative block, additionally equipped with low-pass and washout filters. This scheme takes the local voltage as an input signal, and the POD controller output is fed into the reactive current control of the inverter. The control diagram is depicted in Figure 6, and the switch is in a position to take over the control of $i_{q,ref}$. The goal of the implemented control is to increase the rate of power dissipation in the system, which is the fundamental requirement for the stability of any physical system. The usefulness of the idea was confirmed with simulations, as well as physical experiments.

Singh et al. developed a simple damping controller for both wind turbines and PV generators [91]. It consisted of wash-out and notch filters and a proportional gain. The notch was tuned for the interarea mode frequency. Since the output of the damping controller modulates the active power of the inverter, the gain must be low enough to comply with the equipment ratings. The input signal is the real power measured at the PCC to the power system. This is the first research study that we encountered in our review which examined the electromechanical damping in the frequency domain.

In [92], POD was simultaneously applied to both the active and reactive power control (Figure 7). The active power POD was connected to the battery control, and its feedback signal was the rotor speed deviation of the adjacent synchronous generator. The damped reactive power used the local voltage deviations as feedback. The classical lead-lag structure was used for both the PODs. Particle swarm optimization was used to minimize the transient response-based objective. It is difficult to evaluate whether two PODs significantly outperformed a single damper attached to either the active or the reactive power control, as the authors did not present such an analysis. Hence, the benefits of using two PODs remain unsupported by evidence.



Figure 7. POD modulating the active and/or reactive power.

Li et al. recently published fundamental research on the mechanism of interaction between PV generation and synchronous machines [48]. The authors used a simple SMIB system including a PV generator to analytically derive the inverter current contribution to the damping and synchronizing coefficients acting on the synchronous machine's rotor. The study demonstrated that using rotor angle deviation as a feedback signal stacked with a PI damping controller can affect both the damping and the synchronizing coefficients. Using the frequency deviation as feedback with the PID damper affects both the damping and the synchronizing coefficients, as well as the resultant rotor inertia time constant. The computations indicated that the stabilization is more effective when applied to the inverter's P-control, but it requires the suboptimal operation of the MPPT to provide a sufficient range of active power modulation.

In another study, Silva-Saravia et al. proposed a method called step-down modulation (SDM) that does not require active power curtailment to enable the PV inverter to participate in the mode damping [93]. Upon the detection of electromechanical oscillations, the control method only temporarily shifts to the active power operating point to allow its bidirectional damping modulation. This approach is fairly similar to switching the inverter to STATCOM mode, as in [94]. After the mode is damped, the power is gradually restored to its maximum value. A single lead-lag block is used to compensate for the critical mode.

4.2. Damping Control Based on Wide-Area Measurement Signals

A more recent trend in POD research is to use feedback from WAMS. This section presents a chronological review of the advancements in this research field and examines the problems that demand solutions. The WAMS communication technology itself is a very broad area of study, and a comprehensive review is available in [95].

Shah et al. contributed to the robust design of POD synthesis for inverter-based generation. Their controller modulates the bus voltage reference signal, as shown in Figure 8. A minimax linear quadratic gaussian (LQG)-based power oscillation damper was used in [96] to improve the robust stability of the power system. The authors suggested that the features of minimax optimal controllers achieve an acceptable tradeoff between performance and robustness. The tuning procedure is complex and requires the estimation of an uncertainty matrix based on the probable operating conditions of the network. After the optimal stabilizer structure is synthesized, its order is reduced to make it effectively applicable. A similar study on minimax LQG was presented in [97]. In [98,99], Shah et al. demonstrated the possibility of using WAMS as an input to the inverter POD. In this case, the WAMS signal is the real power measured on a power line. The line to be measured is

determined on the basis of the mode observability technique. In [100], a more advanced study, Shah and Lee synthesized a low-order robust damping controller by defining a convex optimization problem, unlike similar contributions by others that rely on more complex nonconvex definitions [101]. Unlike Shah's previous works, this time, the most suitable feedback signal was selected according to the Hankel singular value. Ge et al. proposed a dominant mode ratio as a metric for the selection of the suitable feedback signal from WAMS [102]. Ge's dominant mode ratio identified the rotor speed deviation signal as the most potent in achieving higher damping, and it used a conventional PSS lead-lag structure. However, [102] did not take into account the WAMS time delay, which could be considered as the major drawback of this study.



Figure 8. POD modulating the bus voltage reference.

Maleki and Varma developed a PV-STATCOM control scheme in [87,94]. At night, the inverter is operated purely as POD STATCOM, while, during the day, it operates in a PV mode and only briefly (for a dozen seconds) switches to the STATCOM mode upon detection of system power oscillations. The active power generation is entirely interrupted during the POD action, following which it is slowly restored to the pre-disturbance conditions. The feedback signal is the current measured on a line somewhere in the power system. In one of the test cases, the authors selected the current of a line that was not near the PCC. This would require the WAMS to provide the feedback signal. This aspect was not considered in these papers. The authors proposed an oscillation detection unit that activates the POD and the STATCOM mode. The POD is a first-order classic lead-lag with a washout filter. The first-order compensation is sufficient to damp a single interarea mode; however, in a real-life setting, there exist many modes. These studies do not provide clear observations concerning how the POD will react to electromechanical modes with frequencies differing from the tuned one, and they do not consider whether it would induce negative damping. The simplex method was used along with a master-slave technique to optimize the damping controller. In 2019, Verma et al. upgraded their PV-STATCOM concept with a fast frequency response to support the grid by reducing the frequency deviations through the inverter active power control [103].

Similar to the concepts already applied to wind power generators, Shen et al. applied goal representation heuristic dynamic programming (GrHDP) as an adaptive POD for PV generators, based on WAMS measurements [104]. The authors employed reactive power modulation for the damping control (see Figure 7). Adaptive dynamic programming was used to compensate for the feedback signal time delay. A time delay up to 750 ms was successfully tested. The latter was measured through the timestamps from the GPS system of the PMU. The GrHDP itself is a neural network (NN)-based controller. The three NNs work together: an Action NN performs the control, a Critic NN assesses the control quality and modifies the Action network accordingly, and a Goal NN modifies the Critic network by generating an internal reward signal. The authors proposed an offline training of the NNs, following which online training can be applied to adapt the POD to the varying operating conditions of the power system. Online training has a major advantage over classical POD control because the power system model details are usually not available to inverter manufacturers. However, extensive additional research is needed to address the issue related to the initial NN weights preset by the manufacturer before the

online training starts following the inverter grid coupling. The problem is that these preset weights can potentially degrade the power system stability, especially if the power plant has a high capacity.

Surinkaew et al. proposed a centralized damping controller based on WAMS [105,106]. The WAMS measures the real power flowing through the tie-lines between the areas. The centralized damping controller sends its stabilizing signals to the conventional generating units, wind generators, and PV inverters. The centralized POD is connected as depicted in Figure 8. The communication time delay is compensated for by a phase-lead block. However, this approach is less sophisticated as compared to others [104], and it is prone to performance degradation because, in real instances, the communication delay is subject to variations; hence, this fixed compensation is not expected to operate robustly. The authors proposed the implementation of a switch between the centralized and the local POD when the communication is interrupted. In a 2019 paper, Surinkaew and Ngamroo addressed the issue of the variable communication time delay of the WAMS by incorporating uncertainty into the POD parameter optimization [107]. In more recent studies, the authors applied a probabilistic selection of the WAMS signal on the basis of combined controllability and observability criteria [108,109]. It was demonstrated that, while a signal can be a good candidate for some operating points, for others, the same signal could be a bad choice. Therefore, the authors proposed the signal selection based on the power system state probability. Byrne et al. employed the vector Lyapunov technique to assess the effect of the structured uncertainties of communications in power systems with increased dispersed generation to improve the distributed control [110].

Multiple-model adaptive POD was developed in [111]. The authors proposed the use of the K-means clustering technique to reduce the number of operating conditions in terms of the observed electromechanical modes. The technique was used to develop a reduced number of damping control models. Each model was optimized for its cluster. During realtime operation, the different models are weighted dynamically, depending on the current operating mode to produce an optimal damping signal. A geometric approach was adopted to select the best feedback signal; the generator rotor angle was the best candidate in this case. This would require a WAMS. The authors tested the robust performance considering the feedback signal time delay but only for values up to 200 ms. No special means were utilized to compensate for the delay, and it seems that the multi-model approach inherently tolerates the latter.

Utilizing WAMS as a feedback signal for POD faces two main practical issues: communication delay and signal availability (reliability). Both of these issues are already being addressed in the literature. Different techniques for the selection of proper feedback measurement signals have also been developed. The analysis shows that the speed deviation of a synchronous machine and the real power flowing through a transmission line are the best stabilizing candidates, which are sometimes not even measured near the PV coupling point. It is even more complicated because it has been demonstrated that the efficiency of the feedback signals changes with the power flow conditions of the grid. Details regarding the practicalities of the general application of wide-area damping control in interconnected systems are available in [112].

4.3. Virtual Synchronous Generators

Zhong et al. proposed a novel static synchronous generator (SSG) control for power inverters [113]. Their idea is to use the well-established theory for synchronous machines and mimic that concerning the control of dispersed generation. This concept requires the presence of a small battery that will store/inject virtual kinetic energy, thus imitating an inertial response. A generalized diagram of SSG control is presented in Figure 9. The rotor mechanical equations block takes the reference active power P_{ref} , the nominal rotor speed ω_n (usually 1 p.u.), and the electromagnetic torque T_e as the inputs. The rotor speed ω_r and angle θ outputs are supplied to the electromagnetic equations block. The inputs of the latter are the measured inverter output currents i_{abc} and a field current component i_f , which is proportional to the field winding mutual inductance M_f . Its outputs are the reactive power Q and the reference three-phase instantaneous voltage $u_{abc,ref}$ used to produce the PWM.



Figure 9. SSG/VSG inverter damping control.

This concept was applied by others for the control of storage systems with the capability of providing kinetic energy to the grid [114] or for hybrid solar power plants [83]. In [115,116], the same concept was named virtual synchronous generator (VSG) and was applied to a distributed generator with storage. Its main goal is to improve electromechanical damping. The PLL, together with the state-of-charge signals, is fed into the VSG control. The latter implements a linearized form of the electromechanical equations that are used to generate virtual inertia. The VSG control modulates the phase angle of the inverter control, i.e., its active power output. The presented simulation results look promising. Similar observations were also obtained from the laboratory experiments with real hardware. Further research by Miura and his team demonstrated an interesting approach by upgrading the VSG control with an alternating moment of inertia [117]. As the frequency tends to converge to the nominal value, the moment of inertia is decreased to speed up the recovery. When the frequency diverges from the nominal value, the moment of inertia is increased to limit the frequency excursion. In [118], the alternating moment of inertia was successfully demonstrated in a power system with multiple VSGs. Similar to the alternating moment of inertia, an alternating (adaptive) damping coefficient was proposed in [119]. The authors of [120] posited an alternative VSG control based on a state-space model using pole placement technique, which eliminates the need for PLL measurement. The results showed that this model provides the same damping characteristics as [116]. VSGs, being similar to actual synchronous machines, have an intrinsic oscillatory mode. Fixed-parameter damping methods using state feedback control were developed in [121] to damp the intrinsic mode. Four of the major VSG control schemes were incorporated in a unified modeling framework that can be used to study the inverter-based generation in different settings: grid-connected mode, islanded single inverter mode and islanded multi inverter mode [122]. A modification of the traditional VSG to improve the converter control under unbalanced voltage conditions was presented in [123].

The classical implementation of the VSG concept was verified for a bulk power system setting in [124]. The authors called the VSG a synchronous power controller. A more comprehensive study on the sensitivity of the power system stability to the VSG parameters was conducted in [125]. The small-signal model that was developed identified that three oscillation modes exist with poor damping induced by (i) the power control, (ii) the virtual impedance control, and (iii) the LC harmonic filter. High power droop gain or mechanical time constant can excite an unstable subsynchronous oscillation. The

value of the virtual resistance can also be a source of instability; low values will induce high-frequency synchronous oscillations (also studied in [126]), while high values will induce subsynchronous ones. The oscillation mode originating from the power control has relatively low damping for inverters connected to low-voltage grids. On the other hand, the LC filter mode dominates for high-voltage connections. Liu et al. also studied the impact of both virtual resistance and inductance in [127]. They concluded that reducing the inductance improves the interarea mode damping. The virtual resistance value depends on the damping strategy. Decreased values may improve the damping through the VSG active power control loop, while increased resistance could be favorable for damping from the VSG voltage control strategy.

The VSG small-signal model was extended in [54] to study the impact of the voltage closed-loop control (VCLC) on the observed oscillation modes. It was found that the VCLC could improve the damping induced by the LC filter, but this may degrade the virtual impedance mode. The VCLC also has a positive impact on the subsynchronous mode when applied to inverters connected to weak grids, such as the low-voltage ones. Another benefit of the VCLC is that it allows the accommodation of higher virtual inertia and droop, thus improving the inverter's inertial response.

In recent years, the VSG concept has been extensively developed, and this development continues to the present day. Although promising, such control has the inherent weaknesses of conventional synchronous machines. They introduce their oscillation modes that have to be dealt with. The main goal of VSG performance optimization is to select proper values for the virtual inertia and impedance, such that the POD control stays beneficial. This is not a straightforward task because the acceptable range or values strongly depend on the grid topology and the inverter's control scheme. General recommendations or a universal methodology for the proper selection of the virtual parameters are not available; hence, this must be done in a case-specific manner.

5. POD in Microgrids through PV Systems

In recent years, microgrids have become a much-promoted concept. A lot of research has been dedicated to developing their concept of operation and economics. Although microgrids are not attributed as much focus as bulk interconnected systems in terms of oscillation damping, some studies demonstrated the possible stability issues that arise from the interactions between power converters and loads and proposed remedial control solutions.

Kazemlou et al. considered an interconnected DC microgrid in [128]. Using only local measurements, the authors developed an adaptive nonlinear controller to mitigate the voltage and power oscillations, based on NN. As it is adaptive, the network learning is online, which omits the need for extensive offline learning with a big dataset. This is a major advantage as a dataset for such systems is not readily available.

Another significant study contributing to the POD control of DC microgrids was presented in [129]. Hamzeh et al. considered a composite generation setting that included a fuel cell, PV system, and supercapacitor. In this type of system, low-frequency oscillations originate from the power-sharing control unit. The control strategy presented in this study comprises a multi-loop voltage controller and a virtual impedance loop as an alternative to the conventional droop control. The virtual impedance loop implements a dynamic droop for efficient damping of the low-frequency oscillations. The authors applied the Guardian map theorem to guarantee a sufficiently robust stability margin. In addition to numerical studies, this paper presented hardware-in-the-loop (HIL) simulations to verify the efficacy of the proposed control. The results suggested that the HIL conveniently facilitates the development of microgrids in laboratory settings.

Jusoh et al. studied a DC microgrid setup consisting of PV arrays and constant power loads [130]. The authors explained the possible detrimental interaction between the power source and the load in terms of stability. The constant power load may present itself as a negative impedance, thus causing stability issues. The authors proposed stabilization of the DC grid using an active damping network that would introduce positive impedance, which would compensate for the negative impedance of the loads. The damping network is a bidirectional DC–DC buck-boost converter, which draws a current proportional to the DC voltage, thus resulting in positive impedance.

Rafi et al. developed a modified VSI control that can operate and smoothly switch between the grid-tied mode and the island mode [131]. In the island mode, the inverter controls the voltage magnitude and frequency. The proposed modification of the typical control design presented robust operation under severe load conditions and improved oscillation damping. Small-signal analysis was used to set the virtual impedance gain to a suitable value.

A quasi-master–slave voltage/frequency control for microgrids was developed in [132]. The proposed control is capable of operating the microgrid both in the islanded mode and in the grid-tie mode. The small-signal stability of the grid was improved through the optimization of the DC capacitance, droop coefficients, and control parameters.

Satapathy et al. proposed the improvement of microgrid stability through an optimization of the converter's PLL and the diesel generator's governor gains, without implementing a dedicated POD [133]. The optimization was performed by developing a variation of the firefly algorithm. An improved firefly stabilization algorithm was developed in [134].

An interconnection between multiple microgrids was considered in [135], where load step changes in the microgrids were demonstrated to induce low-frequency intermicrogrid oscillations. The origin of these oscillations is the primary energy source of the PV installations and the DC voltage controllers. To dampen these oscillations, the authors proposed a POD that acts on both the real and the reactive power controllers. The inputs to the POD are the real and reactive power tie-line exchange errors, as well as the voltage and frequency deviations.

The effect of VSG control in microgrids was studied in [136]. Both the rate of change of frequency (RoCoF) and the frequency nadir were improved during large load disturbances. This study demonstrated that the VSG concept may be beneficial to power systems of any scale.

It is evident that, unlike large power systems, the possible configurations of microgrids are quite diverse, including DC and AC microgrids, independent or interconnected ones, and grids with or without synchronous machines. The nature of the stability issues also varies across the different configurations. The lack of a standard structure creates a major challenge for developing a universal control and damping management. This necessitates tailor-made solutions for every specific microgrid topology. This is the reason why the references cited above were only focused on particular configurations. While the application of firefly algorithms and NNs is a good scientific exercise, it is still far from being a practical reality for engineers. Hence, the development of a more general unified framework for the design and control of microgrids is imperative. The lack of such a framework is a major obstacle to the engineering and mass implementation of microgrids.

6. Summary of the Features of the POD Controllers

This section summarizes the features of the reviewed POD topologies concerning the type and origin of their inputs, destination of the output signals, control strategy, and optimization techniques.

Table 1 presents the types and the sources of the feedback signals utilized for damping control in the various research studies. In most cases, the POD relies on a single input signal, but there are instances where the damper topology requires two or even four signals (in a microgrid application). The signals can be categorized into two groups depending on their source: local and wide-area measurements. Among the local signals, the inverter's active power ΔP is the most used, followed by the frequency deviations Δf . In the studies dedicated to WAMS-based POD, the authors applied different techniques to determine the best signal. In most cases, ΔP was again revealed to be the best candidate, measured either on a tie-line or on a heavily loaded one.

| Reference | Measured Signals |
|--------------------------|--|
| [79,80] | ΔP , ΔU : real power and voltage deviations |
| [83,85,91,96,97,100,120] | ΔP : real power deviations |
| [48,81,82,137] | Δf : frequency deviations |
| [84,90,92] | ΔU : voltage deviations |
| [77] | ΔI : current deviations |
| [48,88] | $\Delta\delta$: rotor angle deviations |
| [98,104–106] | P from WAMS on arbitrary line in the system |
| [94] | I from WAMS on arbitrary line in the system |
| [102] | $\Delta \omega$ from WAMS |
| [116,138] | $\Delta \omega$ from inverter's PLL |
| [86,92,139] | $\Delta \omega$ from nearby synchronous machine (WAMS use is not explicitly defined in the papers) |
| [135] | $\Delta\omega, \Delta U, \Delta P, \Delta Q$ |

Table 1. Feedback signals for POD.

Table 2 summarizes the applied POD control strategies. For several decades, the leadlag phase compensation was widely used for the stabilization of synchronous generators. It is an extensively studied research area with several publications. This is probably the reason why the majority of the POD proposals are based on lead-lag control. Another conventional group of strategies relies on the well-known PID control or its P, I, D, and PI derivatives. In three of the references reviewed here, the POD structure was not fixed and was optimally synthesized. Our survey identified only two papers that studied the application of NNs. As of today, NNs have not been popularized for POD. The current trend, however, is the use of the VSG inverter control strategy. The highest number of research papers was dedicated to this type of damping control. This is probably because it simultaneously addresses two issues; it improves damping, and it provides an inertial response.

| Reference | POD Control Strategy |
|----------------------------------|------------------------|
| [48,79,80] | PID |
| [82,91,138,139] | Р |
| [77,84-86,92,94,102,105,106,137] | Lead-lag |
| [88,89] | Ι |
| [48,135] | PI |
| [96,97,100] | Synthesized controller |
| [90] | D or filtered D |
| [54,83,113-127,140] | VSG/SSG |
| [104,128] | NN |

Table 2. POD controller structure.

The possible destinations (targets) of the damping controller outputs are summarized in Table 3. Some of the first and oldest studies modified the amplitude and the angle of the reference output voltage (see Figure 5). The more recent research studies targeted different parts of the decoupled control. Several studies applied POD to the active power control of the inverter. This approach has its drawbacks that derive from the power curtailment required to allow a margin for power modulation. More recent studies proposed strategies to detect stability-threatening events such that the power needs to be curtailed only in those short periods. Due to this complexity and reduced power efficiency, a more significant part of the research is dedicated to modulating the reactive power or reactive current control loops. Only a few papers proposed the simultaneous application of POD to both the active and the reactive power loops (Figure 7). It remains unknown whether this strategy is advantageous, but it is certainly more complex.

Table 3. The POD signal destination.

| Reference | Stabilizing Signal Is Fed into |
|-------------------------------------|--|
| [79–81,85] | Angle and amplitude controller |
| [48,82,92,135,139] | Both active power and reactive power controllers of the inverter |
| [77,84,90,94,96,97,100,102,104–106] | Power factor/reactive power/quadrature current controller |
| [83,86,88,89,91,116,120,137,138] | Active power controller |

Table 4 summarizes whether the active power control is unidirectional or bidirectional. In the majority of cases, the modulation is bidirectional. Although the unidirectional approach is more favorable in terms of energy efficiency of the PV plant and, to a certain extent, is capable of coping with electromechanical oscillations, the bidirectional approach was proven to provide higher damping at the expense of real power output restrictions.

Table 4. Unidirectional/bidirectional POD through the active power control loop.

| Reference | Unidirectional/Bidirectional Control |
|---------------------|---|
| [81,88,89] | Unidirectional |
| [48,82,83,91] | Bidirectional |
| [86,92,116,120,139] | Bidirectional from an energy storage buffer |
| [79,80,85,135,137] | Authors did not comment or unclear from the paper |

Optimizing the parameters of the POD controller is a crucial step as the wrong settings could significantly degrade the stability. Interestingly, a large number of research papers reviewed here did not comment on this important step (see Table 5). About half of these studies provided an analytical solution for the POD parameter computations, while the remainder defined objective functions and applied numerical optimization techniques to solve them.

Table 5. POD optimization techniques.

| Reference | POD Controller Parameter Optimization |
|--------------------------------|---|
| [77,79–81,86,90,91,97,106,139] | Authors did not comment |
| [48,82-84,88,102,116,120,138] | Analytic |
| [96] | Line search |
| [89,92] | Particle swarm optimization |
| [85,137] | Phase compensation |
| [100] | Solving Lyapunov functions and linear matrix inequalities |
| [94] | Simplex method |
| [105,133,134] | Firefly algorithm |
| [135] | Genetic algorithm |
| [104,128] | Offline and/or online NN training |

7. Discussions on the Practical Application of POD Controllers

The POD of conventional generating units through PSS is based on well-studied, robust, and established techniques. Among them, the phase compensation approach [15,141–143] is the most widely recognized. It is widely used in practice, can be more or less easily verified on-site, and is based on well-defined theory. In contrast, this does not seem to be the case with POD for PV inverters. In most cases, the proposed tuning algorithms are very complex and can hardly be applied by utilities. While the PSS is typically tuned with either a SMIB model or a somewhat extended one, the tuning of the inverter's POD requires very detailed modeling of the power system. The latter is often not available to the plant owners, which makes the optimization of the POD controllers of PV plants hardly possible. On the other hand, the TSOs have full static and dynamic models of the grid. However, the PV plants, even the large ones, are several in number and distributed across the different locations of the grid. Having a large number of units makes it hard to tune and coordinate the PODs.

To overcome these issues, we propose two possible practical approaches: *Coordinated approach*

- Standardize and adopt one or at most a few types of POD controllers for an inverterbased generation.
- Divide the transmission grid into a reasonable number of stability zones that are easier to manage in terms of stability analysis.
- Attach one or a few fictitious identical PV generators to each zone.
- Apply methods to find the optimal parameters for the PODs of the fictitious inverters for each zone.
- Utilities (e.g., TSOs) to request all plant owners in a given zone to set their inverters with the parameter values provided by the TSO.

Generalized approach

- Develop and standardize a universal POD structure and methodology that requires no retuning for the particular grid of application.
- Inverter manufacturers must follow the methodology to preset their PODs solely on the basis of the properties and sizes of their devices.
- This approach enables even the small-scale inverters connected to low-voltage grids to
 participate in the power system stabilization. The VSG is probably the best candidate
 among the reviewed concepts if its remaining issues are resolved and a practical
 methodology is established.

The second solution is more elegant, and it reduces the TSO's burden to tune and coordinate a countless number of PV plants. Fundamentally, it is a plug-and-play solution. However, extensive and complex research is required to develop and prototype this solution and prove that it works safely and consistently in practice. Although the first approach is not too easy to implement, it seems to be the more realistic choice, at least for the near future.

8. Conclusions

The impact of inverter-based PV generation on the small-signal stability of utility networks and microgrids has been widely studied, and, in many cases, this impact has been proven to be negative. One of the main reasons is the reduced inertial response in the power systems and control interactions with power converters. This situation calls for the development of dedicated POD control to mitigate the anticipated small-signal stability degradation.

The research efforts toward developing a POD for PV inverters date back to 1987 and continue to date. Many solutions have been proposed such as classical PID control, lead-lag compensation for stabilization based on WAMS and VSG, synthesized robust controllers, fractional-order controllers, and even artificial NNs. An extensive variety of algorithms have been applied to optimize the performance of these POD controls.

Although intensive research efforts have been dedicated to the development of POD control for PV inverters to date, the technology readiness level required for massive application to electric power systems has still not been achieved. The field offers ample opportunities for future research and standardization. The VSG concept is perhaps the best candidate for a future POD strategy as it is mostly based on the well-established theory of synchronously rotating masses. As found by our review, converters acting as VSGs may introduce undesired behavior and interactions between the current injecting devices in the power system, much like the conventional synchronous generators. Hence, more future research is needed to study the possible effects of massive deployment of VSG devices. Another barrier to be overcome in its application is the development of a more practical methodology that would help to robustly set its virtual parameters: inertia and impedance.

The other options for POD, such as PID, lead-lag, or fractional-order controllers, will require more research for the development of a well-established methodology for tuning their parameters. The main challenges are the need for coordinating the POD control of a huge number of converter-based generators and the need to know a detailed model of the hosting grid to be able to do the tuning.

Funding: The work on this paper was carried out within project "Study of the electric power system stability and frequency control at a predominant share of renewable energy generation" grant /JH07/27/15.12.2016 from the Bulgarian National Science Fund. The APC was funded by Technical University of Varna.

Acknowledgments: The authors are grateful to Chavdar Ivanov for the thorough reading and useful feedback on the draft of this paper. He helped a lot to increase the quality of the paper's contents and presentation.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

| CSI | Current-source inverter |
|-------|---|
| FACTS | Flexible alternate current transmission systems |
| GrHDP | Goal representation heuristic dynamic programming |
| HIL | Hardware-in-the-loop |
| HVDC | High-voltage direct current |
| LQG | Linear quadratic gaussian |
| MPPT | Maximum power point tracking |
| NN | Neural network |
| PCC | Point of common coupling |
| PLL | Phase-locked loop |
| POD | Power oscillation damping |
| PSS | Power system stabilizer |
| PV | Photovoltaic |
| PWM | Pulse width modulation |
| RES | Renewable energy sources |
| RoCoF | Rate of change of frequency |
| SDM | Step-down modulation |
| SMIB | Single-machine infinite bus |
| SPEF | Structure-preserving energy function |
| SSG | Static synchronous generator |
| TSO | Transmission system operator |
| VCLC | Voltage closed-loop control |
| VSG | Virtual synchronous generator |
| VSI | Voltage-source inverter |
| WAMS | Wide-area measurement system |

References

- 1. Kundur, P.; Paserba, J.; Ajjarapu, V.; Andersson, G.; Bose, A.; Canizares, C.; Hatziargyriou, N.; Hill, D.; Stankovic, A.; Taylor, C.; et al. Definition and classification of power system stability. *IEEE Trans. Power Syst.* **2004**, *19*, 1387–1401. [CrossRef]
- 2. Kundur, P. Power System Stability and Control; McGraw-Hill: New York, NY, USA, 1994; ISBN 0070635153.
- 3. Schleif, F.R.; White, J.H. Damping for the Northwest-Southwest Tieline Oscillations—An Analog Study. *IEEE Trans. Power Appar. Syst.* **1966**, *PAS-85*, 1239–1247. [CrossRef]
- 4. Technical Background and Recommendations for Defence Plans in the Continental Europe; ENTSO-E: Brussels, Belgium, 2010; pp. 1–85.
- 5. System Disturbances Review of Selected 1996 Electric System Disturbances in North America; North America Electric Reliability Council: Princeton, NJ, USA, 2002.
- 6. Hsu, Y.Y.; Shyue, S.W.; Su, C.C. Low Frequency Oscillations in Longitudinal Power Systems: Experience with Dynamic Stability of Taiwan Power System. *IEEE Trans. Power Syst.* **1987**, *PWRS-2*, 92–100. [CrossRef]
- Gibson, C.M. Programme of System Proving Tests and Operational Experience. *IEE Proc. C (Gener. Transm. Distrib.)* 1988, 135, 255–260. [CrossRef]
- 8. Interconnection Oscillation Analysis, Reliability Assessment; NERC: Atlanta, GA, USA, 2019.
- 9. Bachmann, U.; Erlich, I.; Grebe, E. Analysis of interarea oscillations in the European electric power system in synchronous parallel operation with the Central-European networks. In Proceedings of the International Conference on Electric Power Engineering, PowerTech Budapest 1999, Budapest, Hungary, 29 August–2 September 1999; Institute of Electrical and Electronics Engineers Inc.: Budapest, Hungary, 1999; p. 49. [CrossRef]
- Tor, O.B.; Gencoglu, C.; Yilmaz, O.; Cebeci, M.E.; Guven, A.N. Damping measures against prospective oscillations between Turkish grid and ENTSO-E system. In Proceedings of the 2010 International Conference on Power System Technology: Technological Innovations Making Power Grid Smarter, POWERCON2010, Hangzhou, China, 24–28 October 2010. [CrossRef]
- Grebe, E.; Kabouris, J.; López Barba, S.; Sattinger, W.; Winter, W. Low frequency oscillations in the interconnected system of Continental Europe. In Proceedings of the IEEE PES General Meeting, PES 2010, Minneapolis, MN, USA, 24–28 October 2010. [CrossRef]
- 12. Kurth, M.; Welfonder, E. Oscillation behaviour of the enlarged European power system under deregulated energy market conditions. *Control Eng. Pract.* 2005, *13*, 1525–1536. [CrossRef]
- 13. Giannuzzi, G.; Lauria, D.; Pisani, C.; Villacci, D. Real-time tracking of electromechanical oscillations in ENTSO-e Continental European Synchronous Area. *Int. J. Electr. Power Energy Syst.* **2015**, *64*, 1147–1158. [CrossRef]
- 14. Analysis of CE Inter-Area Oscillations of 1st December 2016; ENTSO-E: Brussels, Belgium, 2017.
- 15. Wang, H.; Du, W. Power Electronics and Power Systems. In *Analysis and Damping Control of Power System Low-Frequency Oscillations;* Springer: Boston, MA, USA, 2016. [CrossRef]
- 16. Soumyadeep, M.; Ramya, R. A Comprehensive Review of Damping of Low Frequency Oscillations in Power Systems. *Int. J. Innov. Technol. Explor. Eng.* **2019**, *8*, 133–138.
- 17. Nayini, S.; Raghutu, R. Coordinated PSS and STATCOM Controller for Damping Low Frequency Oscillations in Power Systems. *IJERT-Int. J. Eng. Res. Technol.* **2014**, *3*, 609–614.
- 18. Milano, F.; Dorfler, F.; Hug, G.; Hill, D.J.; Verbič, G. Foundations and challenges of low-inertia systems (Invited Paper). In Proceedings of the 20th Power Systems Computation Conference, PSCC 2018, Dublin, Ireland, 11–15 June 2018. [CrossRef]
- Crivellaro, A.; Tayyebi, A.; Gavriluta, C.; Groß, D.; Anta, A.; Kupzog, F.; Dörfler, F. Beyond low-inertia systems: Massive integration of grid-forming power converters in transmission grids. In Proceedings of the 2020 IEEE Power & Energy Society General Meeting (PESGM), Montreal, QC, Canada, 2–6 August 2019; pp. 1–5.
- 20. Annual Renewable Energy Constraint and Curtailment Report 2019; EirGrid: Dublin, Ireland, 2020.
- 21. Peng, Q.; Jiang, Q.; Yang, Y.; Liu, T.; Wang, H.; Blaabjerg, F. On the Stability of Power Electronics-Dominated Systems: Challenges and Potential Solutions. *IEEE Trans. Ind. Appl.* **2019**, *55*, 7657–7670. [CrossRef]
- 22. Meegahapola, L.; Sguarezi, A.; Bryant, J.S.; Gu, M.; Conde, D.E.R.; Cunha, R. Power System Stability with Power-Electronic Converter Interfaced Renewable Power Generation: Present Issues and Future Trends. *Energies* **2020**, *13*, 3441. [CrossRef]
- 23. Markovic, U.; Stanojev, O.; Aristidou, P.; Vrettos, E.; Callaway, D.S.; Hug, G. Understanding Small-Signal Stability of Low-Inertia Systems. *IEEE Trans. Power Syst.* 2021, 1. [CrossRef]
- 24. Ratnam, K.S.; Palanisamy, K.; Yang, G. Future low-inertia power systems: Requirements, issues, and solutions—A review. *Renew. Sustain. Energy Rev.* 2020, 124, 109773. [CrossRef]
- 25. Ulbig, A.; Borsche, T.S.; Andersson, G. Impact of low rotational inertia on power system stability and operation. *IFAC Proc. Vol.* **2014**, *19*, 7290–7297. [CrossRef]
- 26. Mehigan, L.; Al Kez, D.; Collins, S.; Foley, A.; Ó'Gallachóir, B.; Deane, P. Renewables in the European power system and the impact on system rotational inertia. *Energy* **2020**, 203, 117776. [CrossRef]
- Kroposki, B.; Johnson, B.; Zhang, Y.; Gevorgian, V.; Denholm, P.; Hodge, B.M.; Hannegan, B. Achieving a 100% Renewable Grid: Operating Electric Power Systems with Extremely High Levels of Variable Renewable Energy. *IEEE Power Energy Mag.* 2017, 15, 61–73. [CrossRef]
- 28. Tielens, P.; Van Hertem, D. The relevance of inertia in power systems. Renew. Sustain. Energy Rev. 2016, 55, 999–1009. [CrossRef]
- 29. Till, J.; You, S.; Liu, Y.; Du, P. Impact of High PV Penetration on Voltage Stability. In Proceedings of the 2020 IEEE/PES Transmission and Distribution Conference and Exposition (T&D), Chicago, IL, USA, 12–15 October 2020; pp. 1–5. [CrossRef]

- Li, H.; Xu, Y.; Adhikari, S.; Rizy, D.T.; Li, F.; Irminger, P. Real and reactive power control of a three-phase single-stage PV system and PV voltage stability. In Proceedings of the 2020 IEEE Power & Energy Society General Meeting (PESGM), San Diego, CA, USA, 22–26 July 2012. [CrossRef]
- Munkhchuluun, E.; Meegahapola, L. Impact of the solar photovoltaic (PV) generation on long-term voltage stability of a power network. In Proceedings of the 2017 IEEE Innovative Smart Grid Technologies—Asia (ISGT-Asia), Auckland, New Zealand, 11 June 2018; pp. 1–6. [CrossRef]
- Rajan, R.; Fernandez, F.M. Impact of Distributed Virtual Inertia from Photovoltaic Sources on Frequency Regulation in Hybrid Power System. In Proceedings of the 2020 2nd International Conference on Innovative Mechanisms for Industry Applications (ICIMIA), Bangalore, India, 5–7 March 2020; pp. 13–18. [CrossRef]
- Dozein, M.G.; Chaspierre, G.; Mancarella, P. Contribution to System Frequency Stability and Resilience from PV Plants: A Closed-loop System Identification Approach. In Proceedings of the 2020 55th International Universities Power Engineering Conference (UPEC), Turin, Italy, 1–4 September 2020. [CrossRef]
- Munkhchuluun, E.; Meegahapola, L.; Vahidnia, A. Reactive power control of PV for improvement of frequency stability of power systems. In Proceedings of the 2020 IEEE Power & Energy Society General Meeting (PESGM), Montreal, QC, Canada, 2–6 August 2020. [CrossRef]
- Munkhchuluun, E.; Meegahapola, L.; Vahidnia, A. Optimal battery sizing for large-scale solar-PV generation to improve frequency stability. In Proceedings of the 2019 29th Australasian Universities Power Engineering Conference (AUPEC), Nadi, Fiji, 26–29 November 2019. [CrossRef]
- Jawad, A.; Naim, S.A.; Saha, C.; Masood, N. Al Frequency stability enhancement of a large-scale PV integrated grid. In Proceedings of the 2020 11th International Conference on Electrical and Computer Engineering (ICECE), Dhaka, Bangladesh, 17–19 December 2020; pp. 290–293. [CrossRef]
- 37. Peng, Q. Coordination of Virtual Inertia Control and Frequency Damping in PV Systems for Optimal Frequency Support. *CPSS Trans. Power Electron. Appl.* **2020**, *5*, 305–316. [CrossRef]
- 38. Sauer, P.; Pai, M. Power System Dynamics and Stability; Prentice Hall: Upper Saddle River, NJ, USA, 1998; ISBN 0-13-678830-0.
- 39. Anderson, P.; Fouad, A. *Power System Control and Stability*, 2nd ed.; Wiley-IEEE Press: Hoboken, NJ, USA, 2003; ISBN 0-471-23862-7.
- Saadatmand, M.; Gharehpetian, G.B.; Kamwa, I.; Siano, P.; Guerrero, J.M.; Alhelou, H.H. A Survey on FOPID Controllers for LFO Damping in Power Systems Using Synchronous Generators, FACTS Devices and Inverter-Based Power Plants. *Energies* 2021, 14, 5983. [CrossRef]
- 41. Warrier, P.; Shah, P. Fractional Order Control of Power Electronic Converters in Industrial Drives and Renewable Energy Systems: A Review. *IEEE Access* **2021**, *9*, 58982–59009. [CrossRef]
- 42. Tleis, N. Modelling of Voltage-Source Inverters, Wind Turbine and Solar Photovoltaic (PV) Generators; Academic Press: Oxford, UK, 2019. [CrossRef]
- 43. Dogga, R.; Pathak, M.K. Recent trends in solar PV inverter topologies. Sol. Energy 2019, 183, 57–73. [CrossRef]
- 44. Gonzalez-Longatt, F.M.; Torres, J. Modelling and Simulation of Power Electronic Converter Dominated Power System in PowerFactory; Springer: Cham, Switzerland, 2021; ISBN 978-3-030-54123-1.
- Azmi, S.A.; Ahmed, K.H.; Finney, S.J.; Williams, B.W. Comparative analysis between voltage and current source inverters in grid-connected application. In Proceedings of the IET Conference Publications, Edinburgh, UK, 6–8 September 2011; Volume 2011, p. 101. [CrossRef]
- 46. Rizqiawan, A.; Hadi, P.; Fujita, G. Development of grid-connected inverter experiment modules for microgrid learning. *Energies* **2019**, *12*, 476. [CrossRef]
- 47. Wang, X.; Yang, Z.; Fan, B.; Xu, W. Control strategy of three-phase photovoltaic inverter under low-voltage ride-through condition. *Math. Probl. Engl.* **2015**, 2015. [CrossRef]
- 48. Li, M.; Xiong, L.; Chai, H.; Xiu, L.; Hao, J. Mechanism of PV Generation System Damping Electromechanical Oscillations. *IEEE Access* 2020, *8*, 135853–135865. [CrossRef]
- Ramteke, R.G.; Patil, U.V. Design and comparative study of filters for multilevel inverter for grid interface. In Proceedings of the 2014 International Conference on Power, Automation and Communication (INPAC), Amravati, India, 6–8 October 2014; pp. 39–44. [CrossRef]
- 50. Husev, O.; Chub, A.; Romero-Cadaval, E.; Roncero-Clemente, C.; Vinnikov, D. Voltage distortion approach for output filter design for off-grid and grid-connected PWM inverters. *J. Power Electron.* **2015**, *15*, 278–287. [CrossRef]
- 51. Lettl, J.; Bauer, J.; Linhart, L. Comparison of different filter types for grid connected inverter. In Proceedings of the Electromagnetics Research Symposium Proceedings, Marrakesh, Morocco, 20–23 March 2011; pp. 1426–1429.
- 52. Clark, K.; Miller, N.W. Modeling of GE Solar Photovoltaic Plants for Grid Studies; GE Energy: Schenectady, NY, USA, 2009.
- 53. WECC Solar Plant Dynamic Modeling Guidelines; WECC: Salt Lake City, UT, USA, 2014.
- 54. Sun, D.; Liu, H.; Song, P.; Wang, X.; Gong, Y.; Cheng, X.; Wang, J. Impact of Voltage Close Loop Control on Small Signal Stability of Virtual Synchronous Generators. In Proceedings of the 2019 IEEE Innovative Smart Grid Technologies—Asia (ISGT Asia), Chengdu, China, 21–24 May 2019; pp. 1623–1627. [CrossRef]
- 55. Guerrero, J.M.; Vasquez, J.C.; Matas, J.; De Vicuña, L.G.; Castilla, M. Hierarchical control of droop-controlled AC and DC microgrids—A general approach toward standardization. *IEEE Trans. Industrial Electron.* **2011**, *58*, 158–172. [CrossRef]

- 56. Delghavi, M.B.; Yazdani, A. A unified control strategy for electronically interfaced distributed energy resources. *IEEE Trans. Power Deliv.* **2012**, 27, 803–812. [CrossRef]
- 57. Hamrouni, N.; Younsi, S.; Jraidi, M. A flexible active and reactive power control strategy of a LV grid connected PV system. *Energy Procedia* **2019**, *162*, 325–338. [CrossRef]
- Zhou, S.; Liu, J.; Zhou, L.; She, H. Cross-coupling and decoupling techniques in the current control of grid-connected voltage source converter. In Proceedings of the 2015 IEEE Applied Power Electronics Conference and Exposition (APEC), Charlotte, NC, USA, 15–19 March 2015; pp. 2821–2827. [CrossRef]
- 59. Guo, X.Q.; Wu, W.Y.; Gu, H.R. Phase-locked loop and synchronization methods for grid interfaced converters: A review. *Prz. Elektrotechniczny* **2011**, *87*, 182–187.
- Du, W.; Wang, H.F.; Dunn, R. Power system small-signal oscillation stability as affected by large-scale PV penetration. In Proceedings of the 2009 International Conference on Sustainable Power Generation and Supply, Nanjing, China, 6–7 April 2009; pp. 1–6. [CrossRef]
- Hussain, M.T.; El-Amin, I.M. A PV controlled system for damping power system oscillations. In Proceedings of the 2011 IEEE PES Conference on Innovative Smart Grid Technologies—Middle East, Jeddah, Saudi Arabia, 17–19 December 2011; pp. 1–5. [CrossRef]
- 62. Liu, H.; Jin, L.; Le, D.; Chowdhury, A.A. Impact of high penetration of solar photovoltaic generation on power system small signal stability. In Proceedings of the 2010 International Conference on Power System Technology, Hangzhou, China, 24–28 October 2010; pp. 1–7. [CrossRef]
- 63. Gurung, S.; Naetiladdanon, S.; Sangswang, A. Impact of large photovoltaic penetration on small signal stability. In Proceedings of the 2016 IEEE Region 10 Conference (TENCON), Singapore, 22–25 November 2016; pp. 646–650. [CrossRef]
- 64. Shah, R.; Mithulananthan, N.; Lee, K.Y. Contribution of PV systems with ultra capacitor energy storage on inter-area oscillation. In Proceedings of the 2011 IEEE Power and Energy Society General Meeting, Detroit, MI, USA, 24–28 July 2011; pp. 1–8. [CrossRef]
- 65. Shah, R.; Mithulananthan, N. A comparison of ultracapacitor, BESS and shunt capacitor on oscillation damping of power system with large-scale PV plants. In Proceedings of the AUPEC 2011, Brisbane, Australia, 25–28 September 2011; pp. 1–6.
- Ravichandran, S.; Dasan, S.G.B.; Devi, R.P.K. Small signal stability analysis of grid connected Photo Voltaic distributed generator system. In Proceedings of the 2011 International Conference on Power and Energy Systems, Chennai, India, 22–24 December 2011; pp. 1–6. [CrossRef]
- 67. Shah, R.; Mithulananthan, N.; Bansal, R.C. Oscillatory stability analysis with high penetrations of large-scale photovoltaic generation. *Energy Convers. Manag.* 2013, 65, 420–429. [CrossRef]
- 68. Domínguez-García, J.L.; Gomis-Bellmunt, O.; Bianchi, F.D.; Sumper, A. Power oscillation damping supported by wind power: A review. *Renew. Sustain. Energy Rev.* 2012, *16*, 4994–5006. [CrossRef]
- 69. Elliott, R.; Byrne, R.; Ellis, A.; Grant, L. Impact of increased photovoltaic generation on inter-area oscillations in the Western North American power system. In Proceedings of the 2014 IEEE PES General Meeting | Conference Exposition, National Harbor, MD, USA, 27–31 July 2014; pp. 1–5. [CrossRef]
- 70. You, S.; Kou, G.; Liu, Y.; Zhang, X.; Cui, Y.; Till, M.J.; Yao, W.; Liu, Y. Impact of High PV Penetration on the Inter-Area Oscillations in the U.S. Eastern Interconnection. *IEEE Access* 2017, *5*, 4361–4369. [CrossRef]
- 71. Kumar, D.S.; Sharma, A.; Srinivasan, D.; Reindl, T. Stability implications of bulk power networks with large scale PVs. *Energy* **2019**, *187*, 115927. [CrossRef]
- 72. Wang, T.; Bi, T.; Wang, H.; Liu, J. Decision tree based online stability assessment scheme for power systems with renewable generations. *CSEE J. Power Energy Syst.* 2015, *1*, 53–61. [CrossRef]
- 73. Setiadi, H.; Krismanto, A.U.; Mithulananthan, N. Influence of BES system on local and inter-area oscillation of power system with high penetration of PV plants. In Proceedings of the 2017 International Conference on Applied System Innovation (ICASI), Sapporo, Japan, 13–17 May 2017; pp. 1–4. [CrossRef]
- 74. Jia, Q.; Yan, G.; Cai, Y.; Li, Y. Small signal stability analysis of paralleled inverters for multiple photovoltaic generation units connected to weak grid. *J. Engl.* 2017, 2015–2020. [CrossRef]
- 75. Jia, Q.; Yan, G.; Cai, Y.; Li, Y.; Zhang, J. Small-signal stability analysis of photovoltaic generation connected to weak AC grid. J. Mod. Power Syst. Clean Energy 2019, 7, 254–267. [CrossRef]
- 76. Adib, A.; Mirafzal, B.; Wang, X.; Blaabjerg, R. On stability of voltage source inverters in weak grids. *IEEE Access* 2018, *6*, 4427–4439. [CrossRef]
- 77. Gurung, S.; Naetiladdanon, S.; Sangswang, A. Impact of photovoltaic penetration on small signal stability considering uncertainties. In Proceedings of the 2017 IEEE Innovative Smart Grid Technologies—Asia (ISGT-Asia), Auckland, New Zealand, 4–7 December 2017; pp. 1–6. [CrossRef]
- Liu, D.; Shi, L.; Ni, Y. The influence of wind/PV generation uncertainties on power system low frequency oscillation. In Proceedings of the 8th Renewable Power Generation Conference (RPG 2019), Shanghai, China, 24–25 October 2019; pp. 1–8. [CrossRef]
- 79. Kalaitzakis, K.C.; Vachtsevanos, G.J. On the Control and Stability of Grid Connected Photovoltaic Sources. *IEEE Trans. Energy Convers.* **1987**, *EC*-2, 556–562. [CrossRef]

- 80. Ghali, F.M.A.; El-Khashab, H.A.; Abdel-Motaleb, M.S. Dynamic stability analysis of PV-injected power into a parallel AC-DC power system. In Proceedings of the 1994 IEEE 1st World Conference on Photovoltaic Energy Conversion—WCPEC (A Joint Conference of PVSC, PVSEC and PSEC), Waikoloa, HI, USA, 5–9 December 1994; Volume 1, pp. 1056–1059. [CrossRef]
- de Leon, F. Boon-Teck Ooi Damping power system oscillations by unidirectional control of alternative power generation plants. In Proceedings of the 2001 IEEE Power Engineering Society Winter Meeting. Conference Proceedings (Cat. No.01CH37194), Columbus, OH, USA, 28 January–1 February. 2001; Volume 2, pp. 747–752. [CrossRef]
- 82. Li, G.; Ruan, S.Y.; Choi, S.S. A photo-voltaic damping system for suppressing power system oscillation. In Proceedings of the 2009 IEEE International Conference on Control and Automation, Christchurch, New Zealand, 9–11 December 2009; pp. 1681–1686. [CrossRef]
- Remon, D.; Cantarellas, A.M.; Martinez-Garcia, J.; Escaño, J.M.; Rodriguez, P. Hybrid solar plant with synchronous power controllers contribution to power system stability. In Proceedings of the 2017 IEEE Energy Conversion Congress and Exposition (ECCE), Cincinnati, OH, USA, 1–5 October 2017; pp. 4069–4076. [CrossRef]
- Dahal, S.; Mithulananthan, N.; Saha, T. An approach to control a photovoltaic generator to damp low frequency oscillations in an emerging distribution system. In Proceedings of the 2011 IEEE Power and Energy Society General Meeting, Detroit, MI, USA, 24–28 July 2011; pp. 1–8. [CrossRef]
- Ge, Y.; Du, W.; Wang, H.F.; Littler, T.; Cao, J. Robustness of photovoltaic system based stabilizer to mitigate inter-area oscillation in a multi-machine power system. In Proceedings of the 2013 48th International Universities' Power Engineering Conference (UPEC), Dublin, Ireland, 2–5 September 2013; pp. 1–5. [CrossRef]
- Pothisoonthorn, S.; Ngamroo, I. Active power modulation control of hybrid PV generator-battery for power swing stabilization. In Proceedings of the 2016 13th International Conference on Electrical Engineering/Electronics, Computer, Telecommunications and Information Technology (ECTI-CON), Chiang Mai, Thailand, 28 June-1 July 2016; pp. 1–6. [CrossRef]
- Maleki, H.; Varma, R.K. Coordinated control of PV solar system as STATCOM (PV-STATCOM) and Power System Stabilizers for power oscillation damping. In Proceedings of the 2016 IEEE Power and Energy Society General Meeting (PESGM), Boston, MA, USA, 17–21 July 2016; pp. 1–5. [CrossRef]
- Khan, M.H.; Abido, M.A. Controller design of PV System to enhance the dynamic stability of smart grid network. In Proceedings of the 2011 IEEE PES Conference on Innovative Smart Grid Technologies—Middle East, Jeddah, Saudi Arabia, 17–19 December 2011; pp. 1–5. [CrossRef]
- Khan, M.H.; Abido, M.A. Optimizing the Controller of PV System to Enhance the Dynamic Stability of Smart Grid Network. In Proceedings of the 2012 Third International Conference on Intelligent Systems Modelling and Simulation, Kota Kinabalu, Malaysia, 8–10 February 2012; pp. 96–101. [CrossRef]
- 90. Wandhare, R.G.; Agarwal, V. Novel stability enhancing control strategy for centralized PV-grid systems for smart grid applications. *IEEE Trans. Smart Grid* 2014, *5*, 1389–1396. [CrossRef]
- Singh, M.; Allen, A.; Muljadi, E.; Gevorgian, V. Oscillation damping: A comparison of wind and photovoltaic power plant capabilities. In Proceedings of the 2014 IEEE Symposium on Power Electronics and Machines for Wind and Water Applications, Milwaukee, WI, USA, 24–26 July 2014; pp. 1–7. [CrossRef]
- 92. Pothisoonthorn, S.; Ngamroo, I.; Kunakorn, A. Power swing and voltage stabilization by PV generator with active and reactive power controls. In Proceedings of the 2017 IEEE Manchester PowerTech, Manchester, UK, 18–22 June 2017; pp. 1–6. [CrossRef]
- 93. Silva-Saravia, H.D.; Pulgar, H.A.; Tolbert, L.M.; Schoenwald, D.A.; Ju, W. Enabling Utility-Scale Solar PV Plants for Electromechanical Oscillation Damping. *IEEE Trans. Sustain. Energy* **2020**, *1*. [CrossRef]
- 94. Varma, R.K.; Maleki, H. PV Solar System Control as STATCOM (PV-STATCOM) for Power Oscillation Damping. *IEEE Trans. Sustain. Energy* **2019**, *10*, 1793–1803. [CrossRef]
- 95. Rafique, Z.; Khalid, H.M.; Muyeen, S.M. Communication Systems in Distributed Generation: A Bibliographical Review and Frameworks. *IEEE Access* 2020, *8*, 207226–207239. [CrossRef]
- Shah, R.; Mithulananathan, N.; Lee, K.Y. Design of robust power oscillation damping controller for large-scale PV plant. In Proceedings of the 2012 IEEE Power and Energy Society General Meeting, San Diego, CA, USA, 22–26 July 2012; pp. 1–8. [CrossRef]
- John, J.; Revathy, M. B Power oscillation damper design using minimax LQG and LQG methods in large scale PV plants. In Proceedings of the 2016 International Conference on Energy Efficient Technologies for Sustainability (ICEETS), Nagercoil, India, 7–8 April 2016; pp. 591–595. [CrossRef]
- 98. Shah, R.; Mithulananthan, N.; Lee, K.Y.; Bansal, R.C. Wide-area measurement signal-based stabiliser for large-scale photovoltaic plants with high variability and uncertainty. *IET Renew. Power Gener.* **2013**, *7*, 614–622. [CrossRef]
- 99. Shah, R.; Mithulananthan, N.; Lee, K.Y. Large-Scale PV Plant With a Robust Controller Considering Power Oscillation Damping. *IEEE Trans. Energy Convers.* 2013, 28, 106–116. [CrossRef]
- Shah, R.; Mithulananthan, N.; Lee, K.Y. Low-order robust damping controller design for large-scale PV power plants. In Proceedings of the 2014 IEEE PES General Meeting | Conference Exposition, National Harbor, MD, USA, 27–31 July 2014; pp. 1–5. [CrossRef]
- 101. Kim, S.J.; Kwon, S.; Moon, Y.H.; Kim, K.H. Low-order robust power system stabilizer for single-machine systems: An LMI approach. In Proceedings of the IECON Proceedings (Industrial Electronics Conference), Paris, France, 6–10 November 2006; pp. 742–747. [CrossRef]

- 102. Ge, J.; Zhao, D.; Ma, J.; Wang, Y.; Du, H.; Qian, M. Wide-area damping controller design of large-scale PV power plants in interconnected power systems. In Proceedings of the 2016 China International Conference on Electricity Distribution (CICED), Xi'an, China, 10–13 August 2016; pp. 1–5. [CrossRef]
- Varma, R.K.; Akbari, M. Simultaneous Fast Frequency Control and Power Oscillation Damping by Utilizing PV Solar System as PV-STATCOM. *IEEE Trans. Sustain. Energy* 2020, 11, 415–425. [CrossRef]
- Shen, Y.; Yao, W.; Wen, J.; He, H. Adaptive wide-area power oscillation damper design for photovoltaic plant considering delay compensation. *IET Gener. Transm. Distrib.* 2017, 11, 4511–4519. [CrossRef]
- 105. Surinkaew, T.; Ngamroo, I. Two-level coordinated controllers for robust inter-area oscillation damping considering impact of local latency. *IET Gener. Transm. Distrib.* 2017, 11, 4520–4530. [CrossRef]
- 106. Surinkaew, T.; Ngamroo, I. Coordinated DFIG Wind Turbines and Solar PV Generators for Inter-area Oscillation Damping. In Proceedings of the 2018 International Power Electronics Conference (IPEC-Niigata 2018 - ECCE Asia), Niigata, Japan, 20–24 May 2018; pp. 2287–2291. [CrossRef]
- 107. Surinkaew, T.; Ngamroo, I. Inter-Area Oscillation Damping Control Design Considering Impact of Variable Latencies. *IEEE Trans. Power Syst.* 2019, 34, 481–493. [CrossRef]
- 108. Surinkaew, T.; Ngamroo, I. Adaptive Signal Selection of Wide-Area Damping Controllers Under Various Operating Conditions. *IEEE Trans. Ind. Informatics* 2018, 14, 639–651. [CrossRef]
- Ngamroo, I. Wide-area damping controllers of wind and solar power using probabilistic signal selection. *IET Renew. Power Gener.* 2019, 13, 1351–1359. [CrossRef]
- Byrne, R.H.; Trudnowski, D.J.; Neely, J.C.; Schoenwald, D.A.; Wilson, D.G.; Rashkin, L.J. Small signal stability analysis and distributed control with communications uncertainty. In Proceedings of the 2016 International Symposium on Power Electronics, Electrical Drives, Automation and Motion (SPEEDAM), Capri, Italy,, 22–24 June 2016; pp. 1284–1291. [CrossRef]
- 111. Zhou, L.; Yu, X.; Li, B.; Zheng, C.; Liu, J.; Liu, Q.; Guo, K. Damping Inter-Area Oscillations With Large-Scale PV Plant by Modified Multiple-Model Adaptive Control Strategy. *IEEE Trans. Sustain. Energy* 2017, *8*, 1629–1636. [CrossRef]
- 112. Zhang, X.; Lu, C.; Liu, S.; Wang, X. A review on wide-area damping control to restrain inter-area low frequency oscillation for large-scale power systems with increasing renewable generation. *Renew. Sustain. Energy Rev.* 2016, 57, 45–58. [CrossRef]
- 113. Zhong, Q.C.; Weiss, G. Static synchronous generators for distributed generation and renewable energy. In Proceedings of the 2009 IEEE/PES Power Systems Conference and Exposition, PSCE 2009, Seattle, WA, USA, 15–18 March 2009. [CrossRef]
- Van Wesenbeeck, M.P.N.; De Haan, S.W.H.; Varela, P.; Visscher, K. Grid tied converter with virtual kinetic storage. In Proceedings of the 2009 IEEE Bucharest PowerTech: Innovative Ideas Toward the Electrical Grid of the Future, Bucharest, Romania, 28 June–2 July 2009. [CrossRef]
- 115. Sakimoto, K.; Miura, Y.; Ise, T. Stabilization of a power system with a distributed generator by a Virtual Synchronous Generator function. In Proceedings of the 8th International Conference on Power Electronics—ECCE Asia: "Green World with Power Electronics", ICPE 2011-ECCE Asia, Jeju, Korea (South), 30 May-3 June 2011; pp. 1498–1505. [CrossRef]
- 116. Shintai, T.; Miura, Y.; Ise, T. Oscillation Damping of a Distributed Generator Using a Virtual Synchronous Generator. *IEEE Trans. Power Deliv.* **2014**, 29, 668–676. [CrossRef]
- 117. Alipoor, J.; Miura, Y.; Ise, T. Power system stabilization using virtual synchronous generator with alternating moment of inertia. *IEEE J. Emerg. Sel. Top. Power Electron.* **2015**, *3*, 451–458. [CrossRef]
- 118. Alipoor, J.; Miura, Y.; Ise, T. Stability assessment and optimization methods for microgrid with multiple VSG units. *IEEE Trans. Smart Grid* **2018**, *9*, 1462–1471. [CrossRef]
- Zheng, T.; Chen, L.; Wang, R.; Li, C.; Mei, S. Adaptive damping control strategy of virtual synchronous generator for frequency oscillation suppression. In Proceedings of the 12th IET International Conference on AC and DC Power Transmission (ACDC 2016), Beijing, China, 28–29 May 2016; pp. 1–5. [CrossRef]
- Liu, J.; Miura, Y.; Ise, T. A Novel Oscillation Damping Method of Virtual Synchronous Generator Control Without PLL Using Pole Placement. In Proceedings of the 2018 International Power Electronics Conference, IPEC-Niigata—ECCE Asia 2018, Institute of Electrical and Electronics Engineers Inc., Niigata, Japan, 20–24 May 2018; pp. 775–781. [CrossRef]
- 121. Liu, J.; Miura, Y.; Ise, T. Fixed-Parameter Damping Methods of Virtual Synchronous Generator Control Using State Feedback. *IEEE Access* 2019, 7, 99177–99190. [CrossRef]
- 122. Liu, J.; Miura, Y.; Bevrani, H.; Ise, T. A Unified Modeling Method of Virtual Synchronous Generator for Multi-Operation-Mode Analyses. *IEEE J. Emerg. Sel. Top. Power Electron.* 2020. [CrossRef]
- Zheng, T.; Chen, L.; Zhang, H.; Chen, T.; Mei, S. Control strategy for suppressing power oscillation of virtual synchronous generator under unbalanced grid voltage. In Proceedings of the International Conference on Renewable Power Generation (RPG 2015), Beijing, China, 17–18 October 2015; pp. 1–5. [CrossRef]
- 124. Remon, D.; Cantarellas, A.M.; Mauricio, J.M.; Rodriguez, P. Power system stability analysis under increasing penetration of photovoltaic power plants with synchronous power controllers. *IET Renew. Power Gener.* 2017, *11*, 733–741. [CrossRef]
- Dawei, S.; Hui, L.; Peng, S.; Shunan, G.; Yamin, S. Stability Analysis of Photovoltaic Virtual Synchronous Generators. In Proceedings of the 2018 China International Conference on Electricity Distribution (CICED), Tianjin, China, 17–19 September 2018; pp. 1938–1944. [CrossRef]

- 126. Libin, Y.; Hao, Y.; Chunlai, L.; Tai, Z.; Qian, H. Research on Synchronous Frequency Oscillation Mechanism of Photovoltaic Virtual Inverter Under the Coupling of Machine Network. In Proceedings of the 2018 3rd International Conference on Smart City and Systems Engineering (ICSCSE), Xiamen, China, 29–30 December 2018; pp. 537–540. [CrossRef]
- 127. Liu, H.; Sun, D.; Zhao, F.; Tian, Y.; Song, P.; Cheng, X. The influence of virtual synchronous generators on low frequency oscillations. *CSEE J. Power Energy Syst.* 2020, 1–9. [CrossRef]
- 128. Kazemlou, S.; Mehraeen, S. Decentralized Discrete-Time Adaptive Neural Network Control of Interconnected DC Distribution System. *IEEE Trans. Smart Grid* 2014, 5, 2496–2507. [CrossRef]
- 129. Hamzeh, M.; Ghafouri, M.; Karimi, H.; Sheshyekani, K.; Guerrero, J.M. Power Oscillations Damping in DC Microgrids. *IEEE Trans. Energy Convers.* **2016**, *31*, 970–980. [CrossRef]
- 130. Jusoh, A.; Baamodi, H.; Mekhilef, S. Active damping network in DC distributed power system driven by photovoltaic system. *Sol. Energy* **2013**, *87*, 254–267. [CrossRef]
- Rafi, F.H.M.; Hossain, J.; Lu, J. PV microgrid islanded operation analysis with the designed smart VSI. In Proceedings of the 2015 IEEE 10th Conference on Industrial Electronics and Applications (ICIEA), Auckland, New Zealand, 15–17 June 2015; pp. 632–637. [CrossRef]
- Yang, J.; Yuan, W.; Sun, Y.; Han, H.; Hou, X.; Guerrero, J.M. A novel quasi-master-slave control frame for PV-storage independent microgrid. *Int. J. Electr. Power Energy Syst.* 2018, 97, 262–274. [CrossRef]
- 133. Satapathy, P.; Dhar, S.; Dash, P.K. Stability improvement of PV-BESS diesel generator-based microgrid with a new modified harmony search-based hybrid firefly algorithm. *IET Renew. Power Gener.* **2017**, *11*, 566–577. [CrossRef]
- 134. Nayak, P.; Dash, P.K.; Mishra, S.P. Stability assessment and optimal state-feedback system design for PQ-Vf coordination in a PV/DFIG/DSG based microgrid. *Renew. Energy Focus* **2018**, *27*, 14–32. [CrossRef]
- Zhao, Z.; Yang, P.; Wang, Y.; Xu, Z.; Guerrero, J.M. Dynamic Characteristics Analysis and Stabilization of PV-Based Multiple Microgrid Clusters. *IEEE Trans. Smart Grid* 2019, 10, 805–818. [CrossRef]
- Thomas, V.; Kumaravel, S.; Ashok, S. Reduction of Frequency Oscillations in Solar PV Microgrid using Virtual Synchronous Machine. In Proceedings of the 2019 International Conference on Power Electronics Applications and Technology in Present Energy Scenario (PETPES), Mangalore, India, 29–37 August 2019; pp. 1–5. [CrossRef]
- Jaengarun, P. Comparison of POD Function with Power Controller and Voltage Controller for Photovoltaic Plant. In Proceedings of the 2018 International Electrical Engineering Congress (IEECON), Krabi, Thailand, 7–9 March 2018; pp. 1–4. [CrossRef]
- Xiong, J.; Liu, S.; Wang, X.; Zhu, S.; Zheng, J.; Meng, K. A stability enhancement method for inverter-based distributed generation systems. In Proceedings of the 2015 IEEE Power Energy Society General Meeting, Denver, CO, USA, 26–30 July 2015; pp. 1–4. [CrossRef]
- Zhang, X.; Yang, L.; Zhu, X. Integrated control of photovoltaic-energy storage system for power oscillation damping enhancement. In Proceedings of the 2016 IEEE 8th International Power Electronics and Motion Control Conference (IPEMC-ECCE Asia), Hefei, China, 22–26 May 2016; pp. 1571–1575. [CrossRef]
- Nikam, R.J.; Purkar, R.R. Oscillation Damping Reduction Methodology for Distributed Generator Using VSG. In Proceedings of the 2018 International Conference on Smart Electric Drives and Power System (ICSEDPS), Nagpur, India, 12–13 June 2018; pp. 126–131. [CrossRef]
- 141. Gibbard, M.J.; Pourbeik, P.; Vowles, D.J. *Small-Signal Stability, Control and Dynamic Performance of Power Systems*; University of Adelaide Press: Adelaide, South Australia, 2015. [CrossRef]
- 142. Gibbard, M.J. Co-ordinated design of multimachine power system stabilisers based on damping torque concepts. In *Proceedings of the IEE Proceedings C (Generation, Transmission and Distribution)*; The Institution of Electrical Engineers: London, UK, 1988; Volume 135. [CrossRef]
- 143. Lam, D.M.; Yee, H. A study of frequency responses of generator electrical torques for power system stabilizer design. *IEEE Trans. Power Syst.* **1998**, *13*, 1136–1142. [CrossRef]